Author's response to reviews

Title: INvestigational Vertebroplasty Efficacy and Safety Trial (INVEST): a controlled trial of percutaneous vertebroplasty.

Authors:

Leigh A Gray (gray.leigh@mayo.edu)
Jeffrey G Jarvik (jarvikj@u.washington.edu)
Patrick J Heagerty (hagerty@u.washington.edu)
William Hollingworth (william.hollingworth@bristol.ac.uk)
Lydia Stout (lstout@u.washington.edu)
Bryan A Comstock (bac4@u.washington.edu)
Judith A Turner (jturner@u.washington.edu)
David F Kallmes (kallmes.david@mayo.edu)

Version: 2 Date: 19 December 2007

Author's response to reviews:

The requested changes have been made to the title by removing the IDE number from the original.

The affiliations for each author has been written in full as requested.

The abstract has been structured into the headers as requested.

The trial number is in the last section of the abstract identified as the "Trial Registration".

The manuscript has the sections in the order requested with the competing interests, contributions, acknowledgements and references. The section for the authors has been renamed to "Authors' contributions".

Acknowledgements have been added.

The funding is listed, the references have been changed to include all authors (no et. al.).